vc/20-Feb-04
Week 07  starting 09 Feb 04

SM18 Operation Meeting (Tuesdays)

Schedule for now  + see Display : 

	Bench
	Name
	Test type
	Temp
	-
	-

	TBC1
	MBBR1044
	START TESTS WED NIGHT?
	 
	 
	 

	TBC2
	SSS008
	FINISH WED Midday ?
	1.9 K
	 
	 

	TBD1
	MBBR1037
	START WHEN D2 finished?
	 
	 
	 

	TBD2
	MBAR2013
	  Sunday's problem?
	1.9 K
	 
	 

	TBE1
	MBBR1030
	NEW One in today ?
	 
	 
	 

	TBE2
	MBAR3030
	ICS1 WP04 HV Test Warm
	 
	 
	 

	TBF1
	MBBL3082
	START WHEN F2 finishes
	 
	 
	 

	TBF2
	MBBR2044
	PT 5.2 1.5 KA Quench HF
	1.9 K
	 
	 


Test Coordinator this week: Stephane ( + Gijs de Rijk);   MTM  Tech. on call:  (16-3602) : 

General Points

· CRYO Issues ( Jean-Paul)

· ICS Leaks ( a lot over weekend: B Poulnais knows; Glue problem)

· Problem on 2013 on Sun night ( Guy)

· SSS & Switch over ( Guy, Nikolay) for Wed . 

· D1 work on CFB ( Lionel) for corrector leads ?

· PLC CU tests etc for Claude Dehavay ?

· Do not forget PT8  [ramp to 12000A & re-ramp at 120A/sec etc]

· The Must List for Magnetic Measurements ( only 4 remain now)

· ICS To be CALLED UNDER ALL CIRCUMSTANCES WHEN REQUIRED ; they work 24 hrs/day , 7 days/week
Tel. Numbers in SMCR Grey folder + Board
Shift Leaders know the latest situation/prediction cryo etc & WILL do so (despite French etc) 

· SIGNING Traveller at the end :  Remove 2A, & already sign for Standby ( if non-conformitites)  or stripping even if shafts in 
Because Cryo Will not break vacuum till shafts are out ( problem Sun night/Mon)

· Ohmic manual M’ments 


New Templates in New Home page ( Mahesh) ; Link there in old

Start saving in same directory as MTF HV files ( in dfs  ) for ease of upload to MTF

· New : CDPT file to be EXCEL  created from SMTMS  ( will include minimal Quench Table
. ( in progress);  
This FILE must be UPLOADED to MTF 

PROBLEMS OF LAST WEEK

DETAILED FAULTS FOLLOW-UP ( Test Coord  + Stephane S)
	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Tuesday_03_M


	TBD1: TRU 3 problem due to RS232 communication
	AB/CO
	
	Solved during the morning 

	Tuesday_03_A


	TBC1: HV test login refused on sunmta19.

Password file missing
	AB/CO
	HV
	Solved

	Tuesday_03_A


	TBD1: Cabling not correct in the mobile rack of ap1 coil signals
	AB/CO
	HV
	Solved

	Tuesday_03_N


	TBD2: TT821 out of order
	MTM
	
	

	Tuesday_03_N


	TBC1: Water: valve EV did not follow the command of the PLCSU
	MTM
	
	Opened manually.

Solved by Federico

	Tuesday_03_N


	TBD2: Helium not flowing in the anticryostat
	MTM
	
	

	Tuesday_03_N


	TBD1:TRU, mobile rack not scanned at proper time. Qloc not recorded during the training quenches,
	MTM
	
	

	Tuesday_03_N


	TBD1:TRU, mobile rack not available in the database. LHC cycle stopped by HWRC
	MTM
	HRWC
	

	Tuesday_03_N


	TBD1:Re-scanning of the missing data not possible because of a network problem. Site not accessible by the mobic
	MTM
	HRWC
	

	Tuesday_03_N


	TBC1: connector D20 not plugged back after HV Trigger sent when we start to power magnet.
	MTM
	
	

	Tuesday_03_N


	TBC1: SPA: HRWC system gave error: while getting equipment list from MTF in the database..
	MTM
	HRWC
	HRWC disabled and SPA re-launched.

	Wednesday_04_A


	TBD1: last quench before warm up ; problem with LF trigger and I-earth signal
	MTM
	LF
	Retried.  Then OK.

	Wednesday_04_A


	EVC cannot be controlled automatically. The valve is broken and is under repair
	MTM
	
	Handle in manual mode..

	Wednesday_04_N


	TBF1: During a quench AQA give error of 1% in tau on one heater. 
	MTM
	AQA
	Maryline contacted. LF signals checked.

OK. Acquisition error.

	Wednesday_04_N


	TBF1: QH YT221 had a lower resistance than the errors 
	MTM
	
	HV test done then OK.

	Tuesday_05_M


	TBD2: Problem when QH charged because permanent trigger present.

Cable 20 V, 18 A not properly connected (socket missing).
	MTM
	
	Maryline called and repaired.

	Tuesday _05_A


	TBD2: Intercon box not working. No led glowing in the panel. CTH sent information to PLCSE that magnet was below 20 K and it was warm.
	MTM
	ACR/AB-CO?
	A.Tovar called for repair

	Tuesday _05_A


	Mobile rack in wrong state when receiving order from TEMA. Switch in FA position and not in Qloc.
	MTM
	
	

	Tuesday _05_N


	TBE2: IAP at cold trigger on potaim cards Vbus ext-int.
	MTM
	
	Solved by Maryline

	Friday _06_M


	TBF2: Control of the rotary swithch : no led was glowing
	MTM
	
	Maryline contacted

	Friday _06_M


	TBF2: MMP launched : problem of amplifier.
	MTM
	MMP
	Aborted.

	Friday _06_N


	TBC1: HRWC scan for HV XMT.MY92 does not exist.
	MTM
	HRWC
	

	Friday _06_N


	TBE1: Leak discovered during leak tests..
	ICS
	
	ICS called.

	Saturday _07_M


	TBC1: Leak on anticryostat 
	?
	
	

	Saturday _07_M


	TBD1: Leak on the below
	ICS
	
	

	Saturday _07_A


	TBE2: IAP final at warm: alarm on anticryostat X2
	MTM
	
	Guy reset the PLC

	Saturday _07_A


	TBE2: CDW2 non conformity

Corrector resistance too high
	MTM
	
	

	Sunday _08_A


	TBE2 : Slow PA during ramp on the fifth training quench. At a current of 5171 A

 The HV test performed between dipole and ground and it failed.
	MTM
	
	Cryo OK. Problem coming from the magnet?

Investigation done Monday. Everything OK.

	Sunday _08_A


	TBE1:Slow PA could not be launched. Cryo problem.
	ACR
	
	

	Monday_09_M


	TBD2: Permanent trigger. Files from a05 to a049 to be erased.
	MTM
	
	Acquisition is kept off

	Monday_09_M


	No PT8 performed for 1068 and 1030, 1053, 2044..

PT8 mandatory in each magnet.
	MTM
	
	

	Monday_09_M


	TBE: cables running from the mobile rack damaged by the ROCLA?
	ICS?
	
	Repaired by P.Legrand


